Author's response to reviews

Title: Impact of HIV vaccination on laboratory diagnosis: cases report

Authors:

Suthon Vongsheree (suthon@dmsc.moph.go.th)
Archawin Rojanawiwat (archawim191@yahoo.com)
Chanchai Chardbanchachai (chadbuncha@hotmail.com)
John Lerwitworapong (john@samarts.com)
Wichuda Kongpromsook (kvichuda@yahoo.com)
Wiroj Paungtubtim (wirojptt@yahoo.com)
Hansa Thaisri (hansat@dmsc.moph.go.th)
Pathom Sawanpanyalert (pathom@loxinfo.co.th)
Pongnuwat Sri-ngam (p_nuwat@hotmail.com)
Silaporn Pithuk (kungfumail@yahoo.com)
Wimala Inunchot (wimalain@yahoo.com)

Version: 3 Date: 12 Jun 2002

PDF covering letter
June 11, 2002.

Subject: Submission of a manuscript
Dear Editors

I would like to submit a case report on “Impact of HIV vaccination on laboratory diagnosis” for your consideration to be published in Bio Med Central (Medicine). This report can be lesson for another developing countries where any phase III vaccine trials may be studied.

I would also ask waivers for “Article processing charges” since the report is from Thailand where the phase III HIV vaccine trial is under investigation. I propose three experts who are well known in AIDS researches in Thailand as my peer reviewers.

1. Professor Dr. Prasert Thongcharoen
   Faculty of Medicine –Siriraj Hospital
   Mahidol University, Thailand
   E-mail address: siptc@mahidol.ac.th

2. Dr. Ruengpung Sutthent
   Faculty of Medicine –Siriraj Hospital
   Mahidol University, Thailand
   E-mail address: sirst@mahidol.ac.th

3. Dr. Kiat Ruxrungtham
   Faculty of Medicine
   Chulalongkorn University, Thailand
   E-mail address: rkiat@chula.ac.th

Hope to hear from you soon, I remain.

Sincerely yours,

Suthon Vongsheere